Author’s response to reviews

Title: Predicting the multi-domain progression of Parkinson's disease: A Bayesian multivariate generalized linear mixed-effect model

Authors:

Ming Wang (mwang@phs.psu.edu)
Zheng Li (zli@phs.psu.edu)
Eun Young Lee (ellee@pennstatehealth.psu.edu)
Mechelle M. Lewis (mlewis5@pennstatehealth.psu.edu)
Lijun Zhang (lzhang6@pennstatehealth.psu.edu)
Nicholas W. Sterling (nsterling@pennstatehealth.psu.edu)
Daymond Wagner (dwagner1@pennstatehealth.psu.edu)
Paul Eslinger (peslinger@pennstatehealth.psu.edu)
Guangwei Du (guangweidu@pennstatehealth.psu.edu)
Xuemei Huang (xuemei@psu.edu)

Version: 2 Date: 07 Aug 2017

Author’s response to reviews:

Please see our response letter in the supplementary files